

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Ortega, Francesc; Edwards, Ryan D.; Hsin, Amy

Article

The economic effects of providing legal status to dreamers

IZA Journal of Labor Policy

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Ortega, Francesc; Edwards, Ryan D.; Hsin, Amy (2019) : The economic effects of providing legal status to dreamers, IZA Journal of Labor Policy, ISSN 2193-9004, Sciendo, Warsaw, Vol. 9, Iss. 1, pp. 1-28, https://doi.org/10.2478/izajolp-2019-0005

This Version is available at: https://hdl.handle.net/10419/222166

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



Francesc Ortega1*, Ryan Edwards2 and Amy Hsin3

The Economic Effects of Providing Legal Status to DREAMers

Abstract

This study quantifies the economic effects of two major immigration policies aimed at legalizing undocumented individuals that entered the United States as children and completed high school: Deferred Action for Childhood Arrivals (DACA) and the DREAM Act. The former offers only temporary legal status to eligible individuals, whereas the latter provides a track to legal permanent residence. Our analysis is based on a general equilibrium model that allows for shifts in participation between work, college, and non-employment. The model is calibrated to account for productivity differences across workers of different skills and documentation status, and a rich pattern of complementarities across different types of workers. We estimate that DACA increased gross domestic product (GDP) by almost 0.02% (about \$3.5 billion), or \$7,454 per legalized worker. Passing the DREAM Act would increase GDP by around 0.08% (or \$15.2 billion), which amounts to an average of \$15,371 for each legalized worker. The larger effects of the DREAM Act stem from the expected larger take-up and the increased incentive to attend college among DREAMers with a high school degree. We also find substantial wage increases for individuals obtaining legal status, particularly those that increase their educational attainment. Because of the small size of the DREAMer population, and their skill distribution, legalization entails negligible effects on the wages of US-born workers.

Current version:	May 31, 2019
Keywords:	DREAMers, immigration, legalization, undocumented
JEL codes:	I20; I26; J13; J1
Corresponding author:	Francesc Ortega

Dina Axelrad Perry Professor of Economics, Queens College. Address: CUNY, 65-30 Kissena Blvd, Flushing, NY 11367, USA. Email: fortega@qc.cuny.edufortega@qc.cuny.edu

² Research Associate, Berkeley Population Center, UC, Berkeley, CA, USA. Email: ryanedw@berkeley.eduryanedw@ berkeley.edu

³ Associate Professor of Sociology, Queens College, CUNY, NY, USA. Email: amy.hsin@qc.cuny.eduamy.hsin@qc.cuny. edu

[©] The Author(s). 2019 Open Access This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. Cite as: Ortega et al. *IZA Journal of Labor Policy* (2019) 9:5.

1 Introduction

It is estimated that 11–12 million undocumented immigrants currently live in the United States. Having entered the country without authorization or overstayed their visas, they cannot legally work and live under the threat of deportation. Yet, undocumented immigrants are responsible for about 3% of gross domestic product (GDP) nationwide and close to double that figure in states like California, Texas, or Nevada (Edwards and Ortega, 2017).

Whether and how this population should be legally incorporated into the country is a source of great political debate. Similar debates have also taken place in many European countries over the last two decades (Vogel et al. 2011, Orrenius and Zavodny 2016, Devillanova et al. 2018, Fasani 2018), typically resulting in amnesties before unauthorized populations grow too large. As a result, the unauthorized population in Europe as a whole has rarely amounted to more than a few million individuals.

In this context, the current situation in the United States stands out. The last major legalization process in the United States occurred nearly three decades ago under the 1986 Immigration Reform and Control Act (IRCA), which granted legal permanent residency to over 3 million undocumented immigrants (Kossoudji and Cobb-Clark 2002, Casarico et al. 2018). In the decades since IRCA's passage, the political climate has shifted rendering a general legalization process politically infeasible. The discussion has moved toward the less ambitious goal of providing legal status to undocumented youth who were brought to the United States as children, commonly known as DREAMers. This population continues to receive widespread public support, with some recent polls indicating that 86% of the American public would like to offer them legal residency.¹

Yet despite continued public support, Congress has failed to pass legislation offering a path to legal status for DREAMers. In 2010, the DREAM Act, bipartisan legislation offering eligible DREAMers pathways to permanent residence, passed the US House of Representatives but failed to pass the US Senate. In response, in June 2012, President Barack Obama enacted the Deferred Action for Childhood Arrivals (DACA) offering undocumented youth who arrived in the country as children reprieve from deportation and renewable 2-year work permits. On September 5, 2017, President Donald Trump rescinded DACA and urged Congress to explore a legislative solution.²

The goal of this paper is to quantify the economic effects of the two most recent immigrant policy reforms aimed at providing legal status to the DREAMer population—DACA and DREAM Act. We report estimates of the effects on GDP, which is a crucial ingredient in the calculation of the net fiscal balance associated with legalizing DREAMers.³ We also report estimates on the wages of documented and undocumented workers along the education–age distribution, which is informative as to the welfare effects of these policies. Our theoretical framework builds on the work of Borjas (2003), Manacorda et al. (2012)

¹ Washington Post - ABC News, September 2017. https://www.washingtonpost.com/page/2010-2019/ WashingtonPost/2017/09/25/National-Politics/Polling/release_491.xml

² To date, two new versions of the DREAM Act have been introduced and await congressional action. In the US Senate, DREAM Act (S.1615) is a bipartisan bill that is co-sponsored by Senate Republicans Lindsey Graham and Jeff Flake and Senate Democrats Chuck Schumer and Dick Durbin. In the US House, DREAM Act (HR.3440) is also a bipartisan bill that is co-sponsored by Republican Ileana Ros-Lehtinen and Democrat Lucille Roybal-Allard.

³ A 2017 https://www.cbo.gov/publication/53410memo by the Congressional Budget Office estimates that passing the DREAM Act would increase the federal deficit by \$2.6B per year over the next 10 years. However, their calculation assumes that legalization would have practically no effect on GDP.

Ottaviano and Peri (2012), and Edwards and Ortega (2017). We develop a general equilibrium model where production is carried out by means of a multi-level constant elasticity of substitution (CES) production function, which allows for productivity differences across workers of different skills and documentation status, and a rich pattern of complementarities across them.

A novel feature of our framework is that we allow for shifts in participation between work, college, and non-employment. This allows us to consider the effects of legalization policy on the college decisions of undocumented youth. Recent empirical studies have argued that DACA led to a substantial increase in the employment rates of DREAMers, driven by shifts from college enrollment into the workforce (Amuedo-Dorantes and Antman 2017) and (Hsin and Ortega 2017) and by shifts from unemployment into employment (Pope 2016). Our analysis incorporates these effects and discusses the participation effects associated with the DREAM Act as well, which differ in the short and long runs.

To calibrate our model, we rely on data from a special extract of the 2012 American Community Survey (ACS) provided by the Center for Migration Studies (2014), which contains a sophisticated imputation for documentation status (Warren 2014), in addition to the usual information on employment, skills, and wages. Importantly, our 2012 baseline data summarize the economic outcomes of DREAMers immediately prior to DACA.⁴ The data show that, on average, undocumented workers earn 19% less than documented workers with the same education and age. This suggests the existence of a large productivity penalty associated with undocumented status.

We use the calibrated model to simulate the effects of DACA and the DREAM Act relative to the baseline data. On account of the empirical evidence establishing that illegal status negatively affects the productivity of undocumented workers through its negative effects on health and labor market opportunities (Leisy 2011, Gonzales 2011, Hainmueller et al. 2017, Hall and Greenman 2015), we assume that gaining legal status increases the productivity of undocumented workers so as to match the level of documented workers with the same age and education level. This assumption is in line with Monras et al. (2017). As a result of a large legalization process in Spain, these authors find that undocumented immigrant workers become close substitutes for similarly skilled natives.

Between its inception and June 2017, almost 800,000 individuals received DACA permits. Based on the actual take-up of the program, our analysis estimates that DACA increased GDP by 0.018% (about \$3.5 billion), or \$7,454 on average per employed DACA recipient. Our analysis also shows that the wages of DACA recipients increased by around 12%, and that native wages were practically unaffected. The latter is driven by the fact that DACA recipients are a very small share of the workforce and, in addition, because of the high school degree requirement, their skill distribution is very similar to that of natives in the same age group.

Turning now to the analysis of the DREAM Act, our data imply that there were 1.65 million undocumented that arrived in the country as children and had completed high school

⁴ This is important because our data do not allow us to distinguish DACA recipients from non-recipients. As a result, data for the period when DACA was already in operation are likely to underestimate the undocumented wage penalty for DREAMers. DACA was approved in June 2012, but very few permits were granted prior to 2013.

(by 2012) and therefore were eligible for legal status.⁵ It is important to note that the overall number of eligible individuals could be as high as 2.93 million if the DREAMers that do not yet have a high school degree obtain one. Our simulations suggest that the DREAM Act would increase GDP by 0.08% (i.e., \$15.2 billion annually), which amounts to \$15,371 per legalized worker. The reasons for the larger effects, compared to DACA, are the expected larger take-up rate and the increase in educational attainment among DREAMers with a high school degree that decide to obtain some college education in order to qualify for the DREAM Act. However, the positive effects on GDP will take several years to materialize. The reason is that, initially, the positive productivity effect of legalization on GDP will be offset by a negative participation effect driven by the return to college of a subset of DREAMers in the workforce. After a few years, these individuals rejoin the workforce with their enhanced skills, resulting in a substantial increase in GDP. Furthermore, our analysis implies that the wages of most of the DREAMers that obtain legal status will increase by at least 15%, although those that decide to obtain some college education will experience an average 52% increase in wages. At the same time, we find that the DREAM Act will have very minimal effects on the wages of native workers, ranging between 0.4% reductions and 0.4% increases.

The rest of the paper is organized as follows. Section 2 contains the literature review. Section 3 describes our data and Section 4 presents our theoretical framework. Section 5 describes the calibration of the model. Our findings are presented in Section 6 (regarding DACA) and Section 7 (regarding the DREAM Act). Final section summarizes our conclusions.

2 Literature Review

A large body of literature has analyzed the labor market effects of immigration. However, the literature on the effects of legalization or the wage penalty associated with unauthorized status is much smaller, and is almost exclusively reduced form, which is an important limitation in terms of simulating the effects of actual policies. In the context of the United States, several studies have documented substantial wage gaps between similarly skilled documented and undocumented workers. For instance, Hall et al. (2010) estimated a 17% wage disparity between documented and undocumented male Mexicans using the Survey of Income and Program Participation. This estimate is highly consistent with the conclusions of studies quantifying the wage effects of obtaining legal status. Two studies that focus on the 1986 IRCA amnesty estimate the wage penalty for being unauthorized to be around 15% (Kossoudji and Cobb-Clark 2002) and (Lozano and Sorensen 2011). Lynch and Oakford (2013) estimated that gaining legal status and citizenship would allow unauthorized immigrants to earn 25% more within 5 years. Orrenius and Zavodny (2015) provide additional evidence of the existence of a wage penalty associated with undocumented status. This study shows that the introduction of E-Verify, a program that allows employers to verify the legal status of employees, led to a reduction on the wages of undocumented workers. Only one study (Lofstrom et al. 2013) found no evidence of improved employment outcomes following legalization, although this was only the case among the least-skilled immigrants.

⁵ We have restricted our sample to individuals older than 17 in year 2012. We also note that we do not have data on criminal records. As a result, some of these individuals may not satisfy the eligibility condition requiring a clean criminal record.

In the recent years, many researchers have turned to the analysis of legalization using European data (Orrenius and Zavodny 2016). Devillanova et al. (2018) focus on the analysis of the effects of the prospect of legal status on employment rates using data for Italy. This issue is important because often times a requirement to become eligible for legalization is having been employed in the country for a period of time. Also analyzing the Italian experience, Fasani (2018) analyzes the effects of legalization on crime. Monras et al. (2017) focus on Spain's 2004 amnesty, which legalized 0.6 million individuals. Their main finding is that legalization led to a net increase in tax revenue of about 4,000 euros per legalized individual. All these studies consider the whole undocumented population, without considering the educational choices of younger unauthorized individuals.

Some recent studies have developed structural frameworks that are useful to analyze the effects of legalization (as well as the effects of deportation). Edwards and Ortega (2017) emphasize the importance of skill and productivity differences across documented and undocumented workers, and calibrate their model using detailed micro-data (Center for Migration Studies 2014). Chassambouli and Peri (2015) analyze the effects of undocumented migration in frictional labor markets, focusing on the effects on vacancies and unemployment. Machado (2017) builds a related framework that emphasizes inter-generational aspects and allows for estimation of the fiscal effects of legalization.

While the existence of documented–undocumented wage gaps has been clearly established, what is less understood is the nature of these gaps. Several authors have provided evidence of detrimental effects of illegality on the labor market opportunities and health of undocumented workers, which point to the existence of an *undocumented productivity penalty*. For example, illegal status has been shown to increase the risk of depression and anxiety among undocumented youth (Leisy 2011, Gonzales 2011, Hainmueller et al. 2017). Other studies have shown how lack of legal work options confine educated undocumented youth into jobs that are not commensurate with their skills (Gonzales 2011, Gleeson and Gonzales 2012, Cho 2017). In addition, Hall and Greenman (2015) find that unauthorized workers are more likely to work in jobs that are physically strenuous and hazardous and receive no compensating differential for working in dangerous work environments.

Our study is also related to a series of recent empirical studies analyzing the effects of DACA on the labor market outcomes and college participation of DREAMers. Pope (2016) and Amuedo-Dorantes and Antman (2017) use data from the ACS and Current Population Survey (CPS), respectively. Both studies find positive effects of DACA on employment, but disagree on the effects on schooling. Amuedo-Dorantes and Antman (2017) find that DACA reduces college enrollment among probable DACA eligible students, whereas Pope (2016) fails to find evidence of an effect on schooling decisions. Hsin and Ortega (2017) use administrative data on students attending a large public university to estimate the effect of DACA on undocumented students' educational outcomes. Their data are unique because they accurately identify legal status. They find that DACA led to a large increase in dropout rates among undocumented college students enrolled at four-year colleges (though not among those attending community college), providing additional confirmation for the findings in Amuedo-Dorantes and Antman (2017).

In a recent study, Kuka et al. (2018) study the effects of DACA on teenagers' human capital investments. They find large increases in high school graduation rates, along with a reduction in teenage pregnancy and increased college attendance (for women).

3 Data

3.1 Sources

Our data are based on the special extract of the ACS for the year 2012 provided by the (Center for Migration Studies 2014), which contains a sophisticated individual-level measure of *imputed undocumented status* (Warren 2014).⁶ The methodology is based on the so-called *resid-ual method*. The starting point is the non-citizen population that arrived in the United States after 1981 because those arrived earlier were able to obtain legal status under the 1986 IRCA law. The first, and main, step is to apply *logical edits*. This entails using the micro-data to identify individuals who are likely to be documented. Specifically, the key variables employed are occupation, citizenship status of immediate relatives, country of birth, receipt of public benefits, and age at entry.⁷ Clearly, while occupation is only useful for employed respondents, the other logical edits apply equally to all individuals regardless of employment status. As a result, the quality of legal status in the CMS imputation for those who are not employed is probably roughly on par with those of the employed. To further narrow the set of likely undocumented individuals, individual observations are then re-weighted to match aggregate estimates on the presence of foreign-born for 145 origin countries that yield independent totals for each US state and for the total undocumented resident population.⁸

Existing estimates of the characteristics of the imputed unauthorized population obtained from the Census, the ACS and the CPS tend to be largely consistent with each other (Warren 2014, Borjas 2016, Pastor and Scoggins 2016). Nevertheless, the broader validity of the imputation is still being analyzed. Assessments remain constrained by lack of large representative surveys that ask legal status.⁹

3.2 Sample definitions and summary statistics

We restrict to the population age 17–70 in the 2012 ACS. We distinguish between documented individuals, defined as those that were born in the United States or born abroad but deemed as likely authorized on the basis of the imputation, and likely unauthorized foreign-born individuals. Among the likely undocumented population, we will concentrate on DREAMers, defined as individuals that arrived in the country before the age of 18 and have obtained a high school diploma (or similar).

We classify individuals as employed, in college, or doing neither of those activities. More specifically, we consider individuals as enrolled in college if they have a high school degree and report that are currently enrolled in school. An individual is considered employed if he stated so in the ACS survey. Last, we define individuals as *non-employed* if they are not employed and not enrolled in school. Table 1 provides a summary of the data. Column 1 shows that our

⁶ First developed by Passel and Clark (1998), the method has continued to evolve in Baker and Rytina (2013), Warren and Warren (2013), Passel and Cohn (2015).

⁷ Workers with certain occupations that require licensing or background checks, such as legal professions, police and fire, some medical professions, are assumed to be authorized, as well as individuals in government or in the military. Anecdotal evidence shows that there are some unauthorized workers in the military. Nevertheless, the size of this group is very small.

⁸ The method also makes adjustments for under-enumeration in the survey on the basis of time of arrival.

⁹ The Survey of Income and Program Participation (SIPP), also a Census product, directly asks respondents about legal status but is roughly one-sixth the size of the ACS. See Van Hook et al. (2015) for a comparison of results based on the SIPP and the ACS.

	All	Docum.	Undoc.	Undoc17	Undoc17	Undoc17
					HSG+	HSG+, 32-
Employed (%)	61	61	68	60	60	57
College (%)	11	11	6	12	22	25
Non-employed (%)	28	28	27	27	18	18
Total count (Mn)	232.43	222.03	10.40	2.93	1.65	1.42
Wage (\$)	20.61	20.82	16.39	14.73	15.94	14.40
as% Pop	100	95.53	4.47	1.26	0.71	0.61
as% Emp	100	95.53	4.99	1.24	0.70	0.57

Table 1	Data Summary
---------	--------------

Notes: The data are based on the 2012 American Community Survey (CMS version). We restrict to individuals aged 17–70. Non-employed means not working and not in college. Total count refers to the estimated number of individuals in each column (in millions). Hourly wages computed on the basis of full-time workers (35 hours minimum worked usually) at year 2012 prices. Column 1 reports on the total population, including documented (US-born or foreign-born) and likely undocumented. Column 2 refers to (likely) documented individuals only. Column 3 refers to the (likely) undocumented individuals. Column 4 reports on unauthorized individuals who arrived in the country at age 17 or younger. Column 5 adds the additional restriction of having a high school diploma or equivalent. Column 6 adds the additional restriction of being less than 32 years old in year 2012.

data accounts for 232.4 million individuals (age 17-70). Among these, 61% were employed, 11% in college, and 28% doing neither of these two activities. Column 2 reports on the documented population, which amounts to 222 million individuals. Column 3 reports on 10.4 million (likely) undocumented individuals. Their employment rate is 68%, 7 percentage points higher than for the documented population, and their college enrollment rate is 6%, 5 percentage points lower than for the documented population. Column 4 restricts the sample to (likely) unauthorized individuals that arrived in the United States at age 17 or younger, which amounts to 2.93 million individuals. About 1.76 million of these are currently employed and the employment and college attendance rates are essentially the same as for the documented population. In column 5, we further restrict to the 1.65 million undocumented individuals with at least a high school diploma (or similar), which corresponds to our main population of interest. The data show an employment rate of 60% (or about 0.99 million employed individuals) and a college attendance rate of 22%. The last column also imposes the condition of being 32 years old or younger in year 2012, which was required to qualify for DACA. In our data, there are 1.42 million potentially DACA-eligible individuals. The table also reports the mean hourly wages of full-time workers for each column. On average, documented workers earn close to \$21, whereas undocumented workers earn roughly 5 dollars less. Naturally, the bulk of the gap is explained by the lower average education and experience of undocumented workers, but not entirely.

It is also interesting to examine the relative size of these groups. In year 2012, undocumented individuals made up for 4.5% of the population, but almost 5% of employment. Undocumented individuals that arrived in the United States prior to age 18 accounted for about 1.25% of both the population and employment. When we further restrict to undocumented individuals that arrived as children and have a high school diploma (or similar), we find that this group accounts for 0.7% of the population and of employment. Because DREAMers are such a small fraction of the population, the effects of gaining legal status on overall GDP will necessarily be relatively small. Importantly, our analysis will distinguish between workers by education and age, besides legal status. Specifically, we define five age groups: (1) 17–26, (2) 27–36, (3) 37–46, (4) 47–56, and (5) 57–70. We also define four groups on the basis of completed education (in year 2012): (1) high school dropouts, (2) individuals with a high school diploma or General Education Diploma (GED), (3) individuals with some college (i.e., at least 1 year of college or an associate's degree), and (4) individuals with a bachelor's degree (and possibly higher degrees as well). On the basis of our definition, there are no high school dropout DREAMers.

We collapse the individual-level data (using the appropriate sample weights) by education, age, and documentation status. The results are summarized in Table 2. The table reports the shares of the column totals. Columns 1–3 refer to the documented population, which can be broken down into 135 million employed individuals, 24 million attending college, and 63 million doing neither of those two activities. Note that by definition, individuals currently

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Edu	Age	Doc	Doc	Doc	U17	U17	U17	U17HSG	U17HSG	U17HSG
		L	С	Ν	L	С	Ν	L	С	Ν
HSD	1	0.02	0	0.1	0.18	0	0.42	0	0	0
HSD	2	0.01	0	0.03	0.18	0	0.16	0	0	0
HSD	3	0.02	0	0.03	0.08	0	0.05	0	0	0
HSD	4	0.02	0	0.04	0	0	0	0	0	0
HSD	5	0.01	0	0.06	0	0	0	0	0	0
HSG	1	0.06	0.21	0.06	0.2	0.32	0.19	0.35	0.32	0.52
HSG	2	0.06	0.02	0.05	0.13	0.02	0.08	0.22	0.02	0.22
HSG	3	0.07	0.01	0.05	0.03	0	0.02	0.06	0	0.05
HSG	4	0.08	0.01	0.07	0	0	0	0	0	0
HSG	5	0.05	0	0.15	0	0	0	0	0	0
SoCo	1	0.06	0.37	0.02	0.09	0.51	0.03	0.17	0.51	0.09
SoCo	2	0.06	0.09	0.03	0.05	0.05	0.02	0.09	0.05	0.06
SoCo	3	0.06	0.04	0.03	0.01	0.01	0.01	0.02	0.01	0.01
SoCo	4	0.06	0.02	0.04	0	0	0	0	0	0
SoCo	5	0.04	0.01	0.08	0	0	0	0	0	0
CoGrad	1	0.03	0.08	0.01	0.02	0.06	0.01	0.04	0.06	0.03
CoGrad	2	0.08	0.07	0.02	0.02	0.03	0.01	0.04	0.03	0.02
CoGrad	3	0.08	0.03	0.03	0	0	0	0.01	0	0
CoGrad	4	0.08	0.02	0.03	0	0	0	0	0	0
CoGrad	5	0.06	0.01	0.09	0	0	0	0	0	0
Total (M)		134.78	24.03	63.23	1.77	0.37	0.8	0.99	0.37	0.29
Total (M)			222.03			2.93			1.65	

 Table 2
 Baseline Data (2012 ACS) on Documented Population and DREAMers. Shares of Column Totals

Notes: The population is restricted to ages 17–70 and is based on the 2012 ACS. Columns 1–3 refer to the documented population (born in the United States or abroad). Columns 4–6 refer to the likely unauthorized individuals that arrived in the country in the age 17 (DREAMers), and columns 7–9 restrict to the subset of DREAMers with a high school diploma (or similar) in 2012. Education levels are defined as (HSD) high school dropouts, (HSG) high school graduates, (SoCo) some college education, and (CoGrad) college graduates. The age groups are defined as (1) 17–26, (2) 27–36, (3) 37–46, (4) 47–56, and (5) 57–70.

enrolled in college cannot be high school dropouts. Turning to DREAMers (columns 4–6), we find that 0.99, 0.37, and 0.29 million individuals were, respectively, employed, enrolled in college, or doing neither of those two. We also note that under our definition there are no DREAMers in age groups 4 (age 47–56) and 5 (age 57–70).

4 Theoretical Framework

Production takes place by means of a constant-returns Cobb–Douglas production function combining capital and labor. We assume that employers have access to a capital rental market at a fixed rental rate \overline{R} . As a result, the capital stock is proportional to labor, which results in a linear relationship between output and labor: Y = BL, where we will refer to *B* as total labor productivity. Below, we describe the labor aggregate in detail. To close the model, we simply impose market clearing conditions on the output market and on all the skill-specific labor markets.

4.1 The labor aggregate

Let us now describe in detail the labor aggregate *L*. We allow workers to differ in education (e = 1,...,E), age (a = 1,...,A), and documentation status (*Doc*, *Undoc*). In total, the number of labor types is given by $2 \times E \times A$. In our preferred specification, we will focus on four education groups (*E* = 4) and five age groups (*A* = 5).

We aggregate all these types of workers by means of a multi-nested CES aggregator, as in Borjas (2003), Manacorda et al. (2012), and Ottaviano and Peri (2012). To construct the labor aggregate, we need data on the number of workers by education, age, and documentation status. We denote the vector of data by *V*. In addition, we need values for an array of worker productivity terms $\Theta = {\theta}$, one for each worker type, and elasticities of substitution across worker types $\Sigma = {\sigma}$. It is helpful to employ the following compact notation to make explicit the inputs needed to compute the labor aggregates $L(V;\Theta,\Sigma)$.

Specifically, the labor aggregate is given by three levels of CES aggregation, with potentially different elasticities of substitution. To maximize comparability with previous studies, we choose the following nesting structure:

$$L_{e,a} = \theta_{e,a}^{Doc} L_{e,a}^{Doc} + L_{e,a}^{Undoc}, \text{ for } e = 1, 2, \dots, E \text{ and } a = 1, s, \dots, A,$$
(1)

$$L_{e} = C(L_{e,a=1}, ..., L_{e,a=A} | \theta_{e,a}, \sigma_{a}), \text{ for } e = 1, 2, ..., E,$$
(2)

$$L = C(L_{e=1}, \dots, L_{e=E} \mid \boldsymbol{\theta}_{e}, \boldsymbol{\sigma}_{e})$$
(3)

In a general form, the CES aggregator across *M* inputs is defined by

$$C(x_{1},x_{2},...,x_{M} | \theta,\sigma) = \left(\theta_{1}x_{1}^{\sigma/(\sigma-1)} + \theta_{2}x_{2}^{\sigma/(\sigma-1)} + ... + \theta_{M}x_{M}^{\sigma/(\sigma-1)}\right)^{\frac{\sigma-1}{\sigma}}.$$
(4)

Implicitly, the equation for $L_{e,a}$ assumes that within an education–age group, documented and undocumented workers are perfect substitutes (as in Borjas (2003)), despite evidence to the contrary (Manacorda et al. 2012, Ottaviano and Peri 2012). This choice is made to keep the framework as simple as possible and has virtually no effect on the estimated GDP effects (as shown in Edwards and Ortega (2017). However, it will tend to exaggerate the effects of changes in the size and skill composition of the immigrant population on natives. Thus, our analysis of wage effects should be interpreted as providing upper bounds for the effects on native wages.¹⁰

The documented–undocumented relative productivity parameters $\{\theta_{e,a}^{Doc}\}$ will play a crucial role in our analysis. In essence, when we simulate the effects of legalization, we endow undocumented workers with the productivity of documented workers with the same age and level of education. Thus, if these relative productivity parameters are larger than one, legalization will entail an increase in the labor aggregate $L_{e,a}$, as well as in the overall amount of labor L. The increase in labor will then trigger an investment response in the same direction to bring the capital–labor ratio and the marginal product of capital back to their initial level.

4.2 Exploitation of undocumented workers

There is plenty of evidence suggesting that the performance of undocumented workers in the labor market is diminished by their lack of legal status. Clear evidence of this is the overqualification phenomenon (Gonzales 2011, Gleeson and Gonzales 2012, Cho 2017), which is probably more widespread among undocumented workers than for immigrants in general. The typical example of over qualification is when a highly educated immigrant, for example, with a college degree, ends up employed in a low-skill occupation. These occupations are characterized by low productivity and, hence, pay low wages. Individuals in this situation will display very low wages given their education levels, which will translate into large documentedundocumented productivity gaps. More specific to the DREAMer population, there is also evidence that the threat of deportation creates anxiety and depression, which are likely to negatively affect the productivity of these workers (Leisy 2011, Gonzales 2011, Hainmueller et al. 2017). Last, undocumented workers are probably subject to a substantial degree of mismatch in their workplaces, reflecting the fact that they cannot obtain a driver's license and are barred from many jobs because of E-Verify or licensing requirements. As a result, they often end up in jobs that are a poor match for their skills, which results in a very low return to their levels of experience and education.

It is also possible that documented–undocumented wage gaps reflect other factors besides productivity gaps. Some studies (Hotchkiss and Quispe-Agnoli 2009, David Brown et al. 2013, Hirsch and Jahn 2015) suggest that undocumented workers are often not paid their full marginal product. Clearly, their bargaining power is diminished by their lack of legal status, and employers can appropriate a larger part of the surplus generated by the employer–employee match. If exploitation of this type is present and we ignore it, observed wages will *underestimate* the productivity of undocumented workers relative to legal immigrants and natives with the same education and experience. This will result in upwardly biased productivity gaps between documented and undocumented immigrant workers, and will lead to upwardly biased estimates of the gains from legalization.

To allow documented–undocumented relative wages to reflect both productivity differences and exploitation, we assume that unauthorized workers are "taxed" at a rate τ_{eq} by

¹⁰ On the contrary, the effects on the wages of legal immigrants will tend to be underestimated. We also note that we allow for different productivity (and therefore wage) levels between documented and documented workers within educationage cells to accommodate this important feature of the data.

employers.¹¹ The net income of undocumented workers in education age group (e, a) is then given by

$$w_{e,a}^{Undoc} = (1 - \tau_{e,a}) MPL_{e,a}^{Undoc},$$
⁽⁵⁾

where MPL stands for the marginal product of labor of that education-age group.

Because of perfect substitution between documented and undocumented immigrants, their relative wage (within an education experience cell) will be given by

$$\frac{w_{e,a}^{Doc}}{w_{e,a}^{Undoc}} = \frac{\theta_{e,a}^{Doc}}{1 - \tau_{e,a}}.$$
(6)

As we shall see below, the data show substantial wage gaps between documented and undocumented workers in the same education–age category. Because the degree of exploitation is not known, we will need to make an identifying assumption to back out the relative productivity terms from the data on relative wages. In our main specification, we will choose the more standard approach of ignoring exploitation and assume that relative productivity equals relative wages but we will also analyze the alternative scenario where there are no productivity differences between documented and undocumented workers with the same observable skills, $\{\theta_{e,a}^{Doc} = 1\}$, and all wage gaps are explained on the basis of exploitation taxes $\{\tau_{e,a}\}$.¹²

5 Calibration

We need to assign values to the parameters of the model: {B, Θ , Σ , τ }. In our calibration, we will consider E = 4 levels of education and A = 5 age groups. We first consider the following values for the elasticities of substitution. Because workers are increasingly more similar in terms of observable skills as we move up the CES layers, it makes sense to consider elasticities of substitution that (weakly) increase as we move from level 1 to level 2: $\sigma_e \leq \sigma_a$. We adopt fairly standard values for these elasticities: (σ_e , σ_a) = (3, 6). These values are fairly uncontroversial (Card and Lemieux 2001, Goldin and Katz 2008).

Next, we turn to the calibration of the productivities by type of labor. For now, we take the stance that documented–undocumented wage gaps (within education–age groups) are the reflection of productivity differences. As discussed earlier, it is well established empirically that lack of legal status negatively affects labor market opportunities and health, with detrimental effects on worker productivity. We follow a sequential process to calibrate productivity terms Θ and to compute the CES aggregates at each level. The process relies crucially on data on relative wages and employment. We use average hourly wages for full-time workers as our measure of income, but measure employment including workers regardless of their usual hours worked.

We begin with level 3, which aggregates documented and undocumented workers in the same age and completed education groups. Under the assumption of no exploitation ($\tau_{e,a} = 0$ for all *e*,*a*), Equation 6 becomes

$$\frac{w_{e,a}^{Doc}}{w_{e,a}^{Undoc}} = \frac{\theta_{e,a}^{Doc}}{1}.$$
(7)

¹¹ For consistency with the rest of the model, we assume that the proceeds of this tax are distributed in a lump-sum manner to all documented workers.

¹² Ortega and Hsin (2018) analyze the sources of the wage gaps between documented and undocumented workers in the United States. Employing a novel identification strategy, they find that the bulk of the wage gap is explained by the reduction in labor productivity associated with lack of legal status, with a small role for employer exploitation. They conclude that the diminished productivity is largely due to occupational barriers that are specific to undocumented workers.

Thus, documented–undocumented relative wages identify the relative productivity terms. It is then straightforward to compute, for each cell (e, a), the CES labor aggregate L_{ac} .

Next, we turn to level 2. For each education level *e*, given the value of σ_a and data on wages and the values for L_{ea} computed in the previous step, we can easily obtain θ_{ea} from

$$\frac{w_{e,a}}{w_{e,l}} = \left(\frac{\theta_{e,a}}{1}\right) \left(\frac{L_{e,a}}{L_{e,l}}\right)^{-1/\sigma_a}, \text{ for } a = 2, ..., A,$$
(8)

where we have normalized $\theta_{e,1} = 1$. Next, we compute aggregate L_e for each *e* using

$$L_{e} = C(L_{e,1}, ..., L_{e,A} | \boldsymbol{\theta}_{e,a}, \sigma_{a}), \text{ for } a = 2, ..., A.$$
(9)

 θ_{e} denotes the vector of relative productivity terms across experience groups with education.

Finally, level 1 relates the relative wages between the two education groups. For each cell *e*, we obtain $\theta_e = (\theta_1, \theta_2, \theta_3, \theta_4)$ from

$$\frac{w_e}{w_1} = \left(\frac{\theta_e}{1}\right) \left(\frac{L_e}{L_1}\right)^{-1/\sigma_e},\tag{10}$$

and compute L using

$$L = C(L_1, \dots, L_4 | \boldsymbol{\theta}_e, \boldsymbol{\sigma}_e).$$
⁽¹¹⁾

At this point, it is helpful to examine the values that we obtain for these parameters, which are collected in Table 3. Column 1 reports the values for the relative productivity terms (under the assumption of no exploitation). The weighted average of the column is 1.22, indicating that documented workers earn about 22% more than undocumented workers with the same observable skills. Under our assumption of no exploitation, this translates into a sizable productivity gap. We also note that there is a great deal of heterogeneity in the size of the undocumented productivity penalty across skill groups. Consider, for instance, age group 2 (27- to 36-year olds). The documented–undocumented relative productivity terms for this age group are 1.18, 1.26, 1.34, and 1, for education levels 1 (high school dropouts), 2 (high school graduates), 3 (an associate's degree or some college), and 4 (college graduates or higher), respectively. These figures show that the highest gaps are for workers with a high school degree and some college education, and the gap is non-existing for college graduates.

Last, we calibrate the term for total aggregate productivity to match GDP in year 2012 by setting $B^{LR} = GDP/L$.

6 The Effects of Temporary Legal Status: DACA

DACA was launched by President Obama in June 2012. Our baseline data are for year 2012, which can be considered the latest pre-DACA period.¹³ Thus, we can interpret our baseline values for the wages of DREAMers as reflecting their wages while working lacking legal status. On the basis of the key eligibility requirements for DACA, our population of interest are likely undocumented individuals that arrived in the United States prior to their 16th birthday,

¹³ Even though DACA was rolled out in 2012, the number of work permits issued was very low until 2013. Only 1,684 applications were approved by the end of 2012 according to the USCIS.

		(1)	(2)	(3)
Edu	Age	$oldsymbol{ heta}_{e,a}^{Doc}$	$oldsymbol{ heta}_{e,a}$	$ heta_{\scriptscriptstyle e}$
HSD	1	1.04	1	
HSD	2	1.18	2.02	
HSD	3	1.26	2.39	1
HSD	4	1.35	2.56	
HSD	5	1.39	2.28	
HSG	1	1.06	1	
HSG	2	1.26	1.87	
HSG	3	1.38	2.31	2.27
HSG	4	1.44	2.59	
HSG	5	1.59	2.19	
SoCo	1	1.04	1	
SoCo	2	1.32	2.27	
SoCo	3	1.46	2.89	2.65
SoCo	4	1.42	3.1	
SoCo	5	1.38	2.64	
CoGrad	1	1	1	
CoGrad	2	1	2.18	
CoGrad	3	1.07	3.09	5.58
CoGrad	4	1.44	3.4	
CoGrad	5	1.6	2.87	
Avg.		1.22		

 Table 3
 Calibration Productivity Terms

Notes: Productivity terms based on the hourly wages of the corresponding full-time workers, assuming no exploitation. Column 1 reports the productivity (wage) of documented workers relative to undocumented workers who arrived as children in the same education and age groups. The average value reported in the last row uses the distribution of undocumented workers arrived as children over education–age groups as weights—the mode is 15% for HSD in age groups 2 and 3. Column 2 reports the productivity of each education–age type of labor, relative to the first age category in each of the education groups. For groups (*CoGrad*, 1) and (*CoGrad*, 2), the average wages for likely undocumented workers are slightly higher than for likely documented workers. We cap these ratios at 1 because they likely reflect the sampling variability due to small cell sizes. Column 3 reports the productivity of each education group relative to the HSD category. The age groups are (1) 17–26, (2) 27–36, (3) 37–46, (4) 47–56, and (5) 57–70.

were younger than 32 years old in 2012, and had a high school diploma (or GED) in that year.¹⁴ According to our data, this population contains 1.4 million individuals, which is fairly close to the 1.3 million estimated by the Migration Policy Institute (2016).¹⁵

6.1 The DACA counterfactual

As of June 2017, slightly less than 800,000 individuals have been granted DACA permits. This amounts to a take-up rate slightly above 0.5. To take this into account, we denote by ϕ the

¹⁴ There were a number of additional eligibility requirements that cannot be measured using our data, such as having a clean criminal record.

 $^{15 \}quad See \ https://www.migrationpolicy.org/programs/data-hub/deferred-action-childhood-arrivals-daca-profiles.$

DACA take-up rate.¹⁶ Lacking evidence against it, for now we assume that the take-up rate is the same across education and age groups within the DREAMer population.

Based on the existing empirical evidence, it appears that DACA had two effects. First, DACA recipients were given work permits presumably allowing them to access the labor market under the same conditions as documented workers. In the model, we will assume that DACA recipients become indistinguishable from documented workers with the same age and education in terms of productivity. Because DREAMers graduated from a US high school, this assumption seems highly plausible. Quantitatively, the key terms in determining the resulting productivity boost are the relative documented–undocumented productivity terms, $\theta_{e,a}^{Doc}$. Second, there is evidence that DACA triggered a participation effect that led to an increase in employment. According to Pope (2016), the additional workers transitioned from unemployment and according to Amuedo-Dorantes and Antman (2017) and (Hsin and Ortega (2017), they dropped out of college in order to work. This participation effect will magnify the effect of DACA on GDP beyond the productivity boost, at least in the short run.¹⁷

To introduce the participation effect into our model, let δ_c be the fraction of DACA recipients that were in college and decided to drop out of school in order to work.¹⁸ Thus, the number of college students that receive DACA and dropped out before graduation to join the workforce is given by $\delta_c \left(\sum_e \Sigma_a \phi C_{e,a}^{Dream} \right)$, where $C_{e,a}^{Dream} \leq C_{e,a}^{Undoc}$ is the number of undocumented individuals of age *a* and education *e* that are enrolled in college and arrived in the country as children.¹⁹ Likewise, let δ_N be the fraction of DACA recipients that were initially non-employed and started working when they received a DACA permit. This results in an increase in employment equal to $\delta_N \left(\sum_a \Sigma_e \phi N_{e,a}^{Dream} \right)$.

We denote the baseline population in the 2012 data by:

 $V = \left\{ L_{e,a}^{Doc}, C_{e,a}^{Doc}, N_{e,a}^{Doc}, L_{e,a}^{Undoc}, C_{e,a}^{Undoc}, N_{e,a}^{Undoc} \right\}.$

The counterfactual undocumented population under DACA is therefore: for each (e, a),

$$\hat{L}_{e,a}^{Undoc} = L_{e,a}^{Undoc} - \phi L_{e,a}^{Dream}$$
$$\hat{C}_{e,a}^{Undoc} = C_{e,a}^{Undoc} - \phi C_{e,a}^{Dream}$$
$$\hat{N}_{e,a}^{Undoc} = N_{e,a}^{Undoc} - \phi N_{e,a}^{Dream}$$

Turning now to the documented population, for each (e, a),

$$\begin{split} \hat{L}_{e,a}^{Doc} &= L_{e,a}^{Doc} + \phi \left(L_{e,a}^{Dream} + \delta_c C_{e,a}^{Dream} + \delta_n N_{e,a}^{Dream} \right) \\ \hat{C}_{e,a}^{Doc} &= C_{e,a}^{Doc} + \phi (1 - \delta_c) C_{e,a}^{Dream} \\ \hat{N}_{e,a}^{Doc} &= N_{e,a}^{Doc} + \phi (1 - \delta_n) N_{e,a}^{Dream} \,. \end{split}$$

¹⁶ Several reasons explain the low participation in the DACA program. First, the application is a lengthy process and undocumented youth may be concerned about disclosing personal information to immigration authorities, including home address, financial information, and bio-metric data. Moreover, DACA grants temporary protection from deportation for eligible youth but offers no protection for parents or siblings who are not eligible. Thus, applying to the program means not only incurring personal risk but also exposing family members.

¹⁷ The long-run effects will depend on whether these individuals eventually return to college and graduate. As of now, we have no empirical evidence regarding whether or not this is the case.

¹⁸ Note that δ_c is *not* the fraction of DREAMers in college, but only the fraction of that group that decided to drop out of college upon receiving temporary legalization.

¹⁹ $C_{e,a}^{Undoc}$ refers to all undocumented individuals enrolled in college (with the corresponding education and age), and $C_{e,a}^{Dream}$ refers only to those that arrived as children.

Note that the overall population is the same in the counterfactual and baseline scenarios. However, there may be an increase in the overall amount of *labor* because of the differential productivity between documented and undocumented workers. A bit of algebra delivers the key expression summarizing the effects of DACA on the labor aggregates: for each (e, a), the increase in labor is given by

$$\hat{L}_{e,a} - L_{e,a} = \phi \Big[\left(\theta_{e,a}^{Doc} - 1 \right) L_{e,a}^{Dream} + \theta_{e,a}^{Doc} \left(\delta_c C_{e,a}^{Dream} + \delta_N N_{e,a}^{Dream} \right) \Big].$$

$$\tag{12}$$

The first term is the productivity boost associated with legalization. The second term is the participation boost because of DREAMers that were initially in college or non-employed and decided to seek employment because of DACA. Aggregation over age and education groups will deliver the overall increase in *L*. Clearly, the documented–undocumented relative productivity terms, $\{\theta_{e,a}^{Doc}\}$, will play a key role in determining the economic effects of DACA. To the extent that these coefficients are larger than one, temporary legalization through DACA will lead to a net increase in the overall amount of labor. Moreover, because of the linear relationship between labor and output, the percent change in labor will translate into an equal percent change in output. Thus

$$G = \left(\frac{\tilde{Y}}{Y_0}\right) = \left(\frac{\tilde{L}_1}{L_0}\right),\tag{13}$$

and we shall calculate dollar amounts for the effect of DACA on GDP using

$$\tilde{Y} - Y_0 = \left(\frac{\tilde{Y}}{Y_0} - 1\right) Y_0 = (G - 1) Y_0.$$
(14)

6.2 Parameters regarding the effects of DACA

Parameter ϕ stands for the DACA take-up rate. According to United States Citizenship and Immigration Services (USCIS), between its inception in 2012 and 2017 (September 30), 798,980 individuals received protection through DACA. We will set ϕ equal to the ratio between the actual number of DACA applications approved (not counting renewals) and the number of DACA eligible individuals according to our dataset (1.42 million as shown in the last column of Table 1). This results in a value of $\phi = 0.56$.

Parameter δ^c is the probability that a DACA recipient who was in college decides to drop out and join the labor market. Hsin and Ortega (2017) estimate that the college dropout rates for DREAMers in college increased by 4 percentage points when DACA was implemented (reaching 7 percentage points in senior colleges). Their data do not identify DACA recipients and therefore they interpret their estimate as an intent-to-treat effect. Therefore, their estimates correspond more closely to $\phi\delta^c = 0.04$. Given the value for ϕ , we therefore pick $\delta^c = 0.07$. Using CPS data, Amuedo-Dorantes and Antman (2017) also found evidence that DACA reduced college attendance among DACA-eligible college students. Their estimates are somewhat larger than those of Hsin and Ortega (2017), but their identification of unauthorized individuals in the data is less accurate, so we base our calibration on the more conservative estimates.

Parameter δ_N is the probability that a DACA recipient who was non-employed, defined as not working and not enrolled in college, successfully seeks employment. According to Pope (2016), DACA increased the probability of employment for DACA-eligible individuals by 4 to 5 percentage points. His estimates suggest that the increase in employment was fueled by an increase in labor force participation and a decrease in unemployment. As before, a conservative

Param	neter values	
ϕ	0.56	DACA take-up rate
δ^c	0.07	Increased prob. of employment for college students
$\delta^{\scriptscriptstyle N}$	0	Increased prob. of employment for "idle" individuals
Ψ	1	DREAM Act take-up rate
γ_2^L	0.50	Increased prob. of college enrollment for employed individuals
γ_2^N	0.50	Increased prob. of college enrollment for "idle" individuals

Table 4 Additional Parameters	5
---------------------------------------	---

Notes: Key parameter values. The scenario more consistent with the estimates by (Pope 2016) is $\delta^c = 0$ and $\delta^N = 0.07$. And in any case we need to have $\delta^c + \delta^N = 0.07$.

	(1)	(2)	(3)	(4)	(5)
	Δ GDP	ΔGDP	Legalized	Legalized	ΔGDP
Scenarios	Pct. change	\$ Billions	All	Employed	\$ per worker
(1) No participation effect	0.0144	2.8	0.79	0.45	6,217
(2) College participants	0.0178	3.5	0.79	0.47	7,454
(3) Non-emp. participants	0.0170	3.3	0.79	0.46	7,181
(4) Universal take-up	0.0289	5.6	1.42	0.83	6,777
(5) Full exploitation	0.0032	0.6	0.79	0.47	1,340

Table 5 Effects of DACA on GDP

Notes: The eligible group consists of likely unauthorized individuals that entered the country younger than 17 with a high school diploma (or equivalent), and were younger than 32 in year 2012, as required by DACA. Columns 3 and 4 report the number of legalized individuals in our simulation, considering only employed (column 4) or also individuals in college or non-employed (column 3). In scenario 1, $(\phi, \delta_c, \delta_N) = (0.56, 0, 0)$. In scenarios 2 and 5, $(\phi, \delta_c, \delta_N) = (0.56, 0.07, 0)$. In scenario 3, $(\phi, \delta_c, \delta_N) = (0.56, 0, 0.07)$. In scenario 4, $(\phi, \delta_c, \delta_N) = (1, 0.07, 0)$. The dollar amounts in column 2 are computed multiplying the pct. change in GDP in column 1 by the latest GDP estimate available—third quarter of 2017. The last column is the ratio of column 2 to column 5.

interpretation of his estimates implies that $\phi \delta^N = 0.04$ and therefore the probability that an actual DACA recipient who was previously non-employed obtains employment is around $\delta^N = 0.07$. However, it is important to keep in mind that we need to avoid duplicating the increase in employment triggered by DACA (by maintaining $\delta^N + \delta^C = 0.07$). The studies above largely agree on the increase in employment generated by DACA, but disagree on whether the newly employed individuals originated from college or from non-employment. Thus, we will consider the two scenarios separately. The top panel of Table 4 summarizes the DACA-specific parameters.

6.3 Results

As explained above, our calibrated model matches several relevant moments about the US economy in year 2012. Specifically, we match overall GDP and the structure of wages and employment in terms of education, age and documentation status. Now, we turn to the results of our simulation. In terms of outcomes, we first quantify the effects of DACA on GDP and later turn to the effects on the wage structure, emphasizing the effects on the wages of the individuals gaining temporary legalization through the DACA program.

6.3.1 Effects of DACA on GDP

It is helpful to consider first the productivity effect, which is the first part of the expression in Equation 12. At the education–age level, this term only depends on the take-up rate in the program (ϕ) and the documented–undocumented relative productivity term ($\theta_{e,a}^{Doc}$). To isolate this effect, we shut down the participation channels ($\delta_c = \delta_N = 0$) when simulating the scenario where DACA permits are distributed in the numbers observed in the data. The results are presented in the first column of Table 5. In this first scenario, the increase in GDP due to DACA amounts to 0.0144%, which amounts to a \$2.8 billion annually, or \$6,217 per DACA recipient in the workforce.²⁰ While this figure is small relative to the US GDP, it is important to keep in mind that DACA recipients are only about 0.3% of the US population.

Scenario 2 is our preferred scenario. In this case, we allow for a participation effect driven by DREAMers that were initially enrolled in college but decided to drop out in order to work when they received DACA, where the intensity of this effect is based on the estimates by Amuedo-Dorantes and Antman (2017) and Hsin and Ortega (2017). In this case, the effect is about 25% higher than in scenario 1, amounting to a 0.0178% increase in GDP corresponding to \$3.5 billion in the aggregate and \$7,454 per employed DACA recipient. In scenario 3, we consider the alternative participation effect based on the estimates by Pope (2016), where the inflow of DREAMers into employment originates in individuals that were previously non-employed. The results imply a slightly smaller GDP gain than in scenario 2, with a GDP increase of 0.0170%, amounting to \$7,181 per employed DACA recipient. It is also worth noting that this increase in GDP is solely due to the effects of legalization. A full assessment of the economic contribution of undocumented workers to the economy needs to take into account the value added of these workers prior to receiving DACA (as in Edwards and Ortega (2017)). We will return to this point in the next section.

Scenario 4 estimates the potential gains from DACA, in the case that all 1.42 million eligible individuals received protection under the program. In this case, the GDP increase could have reached almost 0.03% of GDP. Last, scenario 5 considers an alternative calibration where we assume that the wage gaps between similarly skilled documented and undocumented workers are exclusively due to exploitation. In this case, the calibration entails $\theta_{e,a}^{Doc} = 1$ and we are effectively turning off the productivity effect and are left exclusively with the participation effect. In this case (scenario 5), DACA would have led to a meager 0.0032% increase in GDP. Clearly, the assumption of full exploitation as an explanation of the relative wage gaps between documented and undocumented workers is very extreme, given the extensive empirical evidence in support of the detrimental productivity effects of undocumented status.²¹

6.3.2 Wage effects of DACA

We now turn to the wage effects of DACA. Before discussing the details, it is important to keep in mind that DACA beneficiaries are a very small share of the US population and, as a result, their impact on the wages of natives is bound to be very small. Naturally, the effect on the wages of the DREAMers obtaining legal status will be much larger.

²⁰ Keep in mind that some DACA recipients are in college or non-employed.

²¹ If we had accurate estimates of the degree of exploitation, we would be able to separately calibrate the exploitation tax and the relative productivity terms. However, the existing empirical literature does not offer an estimate of the extent of exploitation for undocumented workers.

Edu	Age	(1)	(2)	(3)	(4)	(5)	(6)
Group	group	Wage growth	Labor	Wage growth	Wage growth	Labor	Wage growth
		Doc-Doc	shares	by edu	Legalized	shares	by edu
		Undoc-Undoc	Doc	Doc-Doc	Undoc-Doc	DREAMers	Legalized
HSD	1	0.01	0.24	0.00		0	
HSD	2	0.01	0.17			0	
HSD	3	0.01	0.20			0	
HSD	4	0.01	0.23			0	
HSD	5	0.01	0.16			0	
HSG	1	-0.03	0.19	-0.01	5.96	0.68	12.43
HSG	2	-0.04	0.18		26.17	0.32	
HSG	3	0	0.21			0	
HSG	4	0	0.26			0	
HSG	5	0	0.16			0	
SoCo	1	-0.02	0.22	-0.01	3.98	0.72	11.73
SoCo	2	-0.02	0.22		31.64	0.28	
SoCo	3	0	0.21			0	
SoCo	4	0	0.22			0	
SoCo	5	0	0.14			0	
CoGrad	1	0	0.10	0.00	0.00	0.53	0.00
CoGrad	2	0.01	0.25		0.00	0.47	
CoGrad	3	0.01	0.25			0	
CoGrad	4	0.01	0.23			0	
CoGrad	5	0.01	0.17			0	

Table 6 Wage Effects of DACA. Percent Changes Relative to Baseline

Notes: We report percent changes in DACA counterfactual relative to baseline. Doc-Doc (Undoc-Undoc) refers to individuals that were documented (undocumented) both in the baseline and in the counterfactual. Legalized individuals are those that had undocumented status in the baseline but were documented in the DACA counterfactual. Because documented and undocumented that do not change legal status with the same education and age are perfect substitutes in production, they experience identical wage growth rates. We use the baseline elasticities (scenario 2 in Table 5). Columns 1 and 4 report wage growth (in percent) by education–age. Columns 2 and 5 report the labor shares in the baseline among documented workers (column 2) and among DREAMers eligible for DACA (column 5). Columns 3 and 6 report age-weighted average wages by education on the basis of the respective previous two columns. A "." denotes a missing value due to the fact that there are no individuals in that education–age-documentation status category. The age groups are (1) 17–26, (2) 27–36, (3) 37–46, (4) 47–56, and (5) 57–70.

The wage effects of our simulation are reported in Table 6. We begin with column 1, which reports the percent change in wages relative to baseline for workers that did not change documentation status, that is, for documented workers or undocumented workers that did not receive DACA permits. Because we assumed that documented and undocumented workers with the same observable skills are perfect substitutes, these two groups experience the same percent change in their wages. Column 1 shows that the wage effects of DACA are negligible. To a large extent, this is due to the change in the relative skill supplies of DACA is very small given the small size of the group of DACA recipients relative to overall employment. The largest effects entail a 0.04% reduction in the wages of high school graduates (age group 2) and a 0.02 percent reduction in the wages of workers with some college (age groups 1 and 2). Column 3 aggregates these figures by education group, weighting each age–education group by their age shares by education (from column 2). The resulting figures

show 0.01% drops in the wages of high school graduates and individuals with some college, and practically zero effects on the wages of workers at the top and bottom of the education distribution.

Column 4 reports the percent changes in the wages of the DACA recipients, which on the basis of the eligibility criteria consisted only of DREAMers with at least a high school diploma in age groups 1 and 2. These individuals experienced a substantial productivity increase. The figures in the table show sizable increases for all age–education groups containing legalized individuals, reaching up to 31%. However, there is a great deal of heterogeneity in the size of the wage growth across education–age groups of legalized individuals. The largest increases pertain to individuals in age group 2 (27- to 36-year olds) with a high school degree or some college. Column 6 provides the corresponding age-weighted averages by education level. The average DACA recipient with a high school degree experienced a 12.43% increase in wages. Likewise, individuals with some college experienced average wage increases of 11.73%. In contrast, we do not find evidence of significant wage growth for the average DACA recipient with a logure of significant wage growth for the average DACA recipient with a college degree. The reason is that the documented–undocumented relative productivity for this group turned out to be essentially 1 in our calibration (see Table 3). Thus, legalization did not improve their labor market outcomes.

7 The Effects of Permanent Legalization

7.1 The DREAM Act counterfactual

According to the 2017 Senate version of the DREAM Act, obtaining permanent residence is a two-stage process.²² The first stage provides eligible individuals with *conditional status*, that is, reprieve from deportation and a work permit. The key requirements for conditional status that can be measured using our data are as follows: (i) having arrived in the country at age 17 or younger and (ii) having graduated from high school or obtained a GED.²³ The second stage of the process imposes additional requirements to obtain legal permanent residence. Eligible individuals must satisfy one of the following criteria by the end of the conditional status period (besides maintaining a clean criminal record): (i) obtaining an associate's degree or at least 2 years of college education toward a bachelor's degree; (ii) 2 years of military service; or (iii) 3 years of continuous employment.

On the basis of the 2014 ACS, the Migration Policy Institute estimates that 1.8 million individuals are eligible for conditional status in year 2017, out of an overall 3.3 million individuals that arrived in the country illegally as children. In comparison, our estimates based on the 2012 ACS for these figures are 1.4 million—undocumented that arrived by the age of 17 and currently hold a high school diploma—and 2.9 million, respectively (Table 1). Our main estimates of the economic effects of the DREAM Act will be based on the 1.4 million individuals already eligible for conditional status.

Individuals that obtain *conditional status* will benefit from relief from deportation and a work permit, much like was the case for DACA recipients. In terms of the model, we will simply consider them as having the same productivity as documented workers with the same age

²² The current version of the House bill has similar requirements, though a little more restrictive.

²³ Like for DACA, a clean criminal record is also a requirement to obtain conditional status.

and education. This is exactly the productivity boost considered earlier, potentially differing only in the take-up rate ψ . Unlike in the case of DACA, we believe that the take-up rate for the DREAM Act will be practically universal among eligible individuals since there is no fear from deportation ($\psi = 1$). We also believe that, unlike DACA, conditional status is unlikely to induce DREAMer college students to drop out. The reason is that their planning horizon remains unchanged and the returns to a college degree will increase thanks to the permanent legal status.

In fact, the requirements to obtain permanent residence in the DREAM Act will likely generate dynamic participation effects. As noted earlier, one of the routes to satisfy the permanent-residence requirement in the second stage is to obtain at least 2 years of college education. This will raise college attendance among individuals in conditional status, relative to what we would have observed otherwise. Thus, unlike for DACA, we may see a *negative* labor market participation effect in the short run when some employed DREAMers quit their jobs to enroll in college. In the long run, these workers will come back to the workforce with some college education and enhanced productivity, which will imply a *positive* participation effect. We believe that this *educational boost* will take place primarily among DREAMers with a high school degree, who are the most likely population to choose the college education route to fulfill the permanent residence requirement.²⁴

7.2 Short-run participation effects

More specifically, we define γ_e^L and γ_e^N to be, respectively, the *increase* in the probability of college enrollment for working and non-employed DREAMers, respectively. Lacking empirical estimates of the size of these effects, we shall assume that these probabilities are zero for individuals with less than a high school education or already having some college education: $\gamma_e^L = \gamma_e^N = 0$ for $e \neq 2$. The reason is that it is much more likely that these individuals will choose to fulfill the permanent residence requirement by joining the Army or being continuously employed for the required number of years. However, many DREAMers with a high school diploma are likely to choose to attend college to fulfill the additional requirement. Thus, we will set $\gamma_2^L \ge 0$ and $\gamma_2^N \ge 0$ in our calibration.

As a result, the short-run counterfactual undocumented population under the DREAM Act is as follows. For each (*e*, *a*), a fraction ψ of all DREAMers (undocumented that arrived in the country as children) receives conditional status:

$$\begin{split} \tilde{L}^{Undoc}_{e,a} &= L^{Undoc}_{e,a} - \psi L^{Dream}_{e,a} \\ \tilde{C}^{Undoc}_{e,a} &= C^{Undoc}_{e,a} - \psi C^{Dream}_{e,a} \\ \tilde{N}^{Undoc}_{e,a} &= N^{Undoc}_{e,a} - \psi N^{Dream}_{e,a} \,. \end{split}$$

Turning now to the documented population, a fraction γ_e^L of the Dreamers in the workforce with education level *e* that received conditional status ($\psi L_{e,a}^{Dream}$) will quit their jobs in order to enroll in college. Likewise, a fraction γ_e^N of the non-employment Dreamers with education

²⁴ Evidence of educational responses by DREAMers to increased returns to education is provided by Kuka et al. (2018). Because these dynamics are driven by the educational incentive built into the DREAM Act, the likely length of the transition phase is 2–4 years. At that point, the long-run effects would materialize.

level *e* that received conditional status ($\psi N_{e,a}^{Dream}$) will enroll in college. More specifically, for each (*e*, *a*),

$$\begin{split} \tilde{L}_{e,a}^{Doc} &= L_{e,a}^{Doc} + \psi \left(1 - \gamma_e^L \right) L_{e,a}^{Dream} \\ \tilde{C}_{e,a}^{Doc} &= C_{e,a}^{Doc} + \psi \left(C_{e,a}^{Dream} + \gamma_e^L L_{e,a}^{Dream} + \gamma_e^N N_{e,a}^{Dream} \right) \\ \tilde{N}_{e,a}^{Doc} &= N_{e,a}^{Doc} + \psi \left(1 - \gamma_e^N \right) N_{e,a}^{Dream}. \end{split}$$

In our calibration, we shall set $\gamma_2^L = \gamma_2^N = \gamma$ for simplicity. We believe that a plausible value for this parameter is $\gamma = 1/2$, that is, one in two DREAMers with a high school degree will choose to obtain some college education in order to qualify for permanent residence. However, we will also produce estimates for higher and lower values of this parameter. Table 4 gathers the key parameter values in the simulation of the effects of the DREAM Act.

7.3 Long-run participation effects

The DREAMers that were initially in the workforce or non-employed in the baseline data but decide to attend college because of the eligibility requirements, $\psi\gamma(L_{2,a}^{Dream} + N_{2,a}^{Dream})$, are now graduating from college with their enhanced skills. We make two conservative assumptions. First, we assume that the DREAMers that went back to school obtain only the *minimum* college education required to satisfy the permanent residence requirement. Namely, those individuals transition from education group 2 (high school graduate) to education group 3 (some college or an associate's degree). Second, we assume that individuals that were initially non-employed stay in that state despite their increased educational attainment.²⁵ Thus, the size of the workforce is unchanged relative to the baseline, although some individuals have upgraded their skills.

The long-run undocumented population under the DREAM Act is the same as it was in the short-run scenario. For each (e, a),

$$\begin{split} \overset{Undoc}{\tilde{L}}_{e,a} &= L_{e,a}^{Undoc} - \psi L_{e,a}^{Dream} \\ \overset{\sim}{\tilde{C}}_{e,a}^{Undoc} &= C_{e,a}^{Undoc} - \psi C_{e,a}^{Dream} \\ \overset{\sim}{\tilde{N}}_{e,a}^{Undoc} &= N_{e,a}^{Undoc} - \psi N_{e,a}^{Dream} \end{split}$$

Turning now to the documented population, for each (e, a), the workforce will be given by

$$\begin{split} \tilde{\bar{L}}_{e,a}^{Doc} &= L_{1,a}^{Doc} + \psi L_{1,a}^{Dream}, & \text{for } e = 1 \\ &= L_{2,a}^{Doc} + \psi \left(1 - \gamma_{2}^{L} \right) L_{2,a}^{Dream}, & \text{for } e = 2 \\ &= L_{3,a}^{Doc} + \psi \left(L_{3,a}^{Dream} + \gamma_{2}^{L} L_{2,a}^{Dream} \right), & \text{for } e = 3 \\ &= L_{4,a}^{Doc} + \psi L_{4,a}^{Dream}, & \text{for } e = 4, \end{split}$$

where the group with some college (e = 3) includes the high school graduates that attended college to fulfill the permanent residence requirement. Importantly, these equations assume that

²⁵ This assumption is probably overly conservative but we are unsure what fraction of these newly minted graduates would ultimately enter the workforce. If all the DREAMers that transition from education level 2 to education level 3 were to become employed, the number of documented individuals with education level 3 in the long-run DREAM Act scenario $(\tilde{L}_{3,a}^{Doc})$ would have to be increased by $\psi \gamma N_{2,a}^{Dream}$. Accordingly, $\tilde{N}_{3,a}^{Doc}$ would have to be decreased by the same amount. Clearly, this would have an additional positive effect on GDP growth.

initially non-employed DREAMers that decided to obtain some college education to qualify for permanent residence remain non-employed in the long-run counterfactual.

As for the non-employed and the college-enrolled population,

$$\begin{split} & \stackrel{\scriptscriptstyle \sim}{\overset{\scriptscriptstyle \sim}{N}} \stackrel{\scriptscriptstyle Doc}{\underset{e,a}{\overset{\scriptscriptstyle >}{e_{e,a}}}} = N^{\scriptscriptstyle Doc}_{e,a} + \psi N^{\scriptscriptstyle Dream}_{e,a} \\ & \stackrel{\scriptscriptstyle \sim}{\overset{\scriptscriptstyle \sim}{C}} \stackrel{\scriptscriptstyle Doc}{\underset{e,a}{\overset{\scriptscriptstyle \sim}{e_{e,a}}}} = C^{\scriptscriptstyle Doc}_{e,a} + \psi C^{\scriptscriptstyle Dream}_{e,a}, \quad for \ all \ e \ and \ a. \end{split}$$

7.4 Results

7.4.1 Effects of the DREAM Act on GDP

Our estimates for the long-run effects on GDP from passing the DREAM Act are reported in Table 7. We consider a variety of scenarios that differ in the value of the parameter governing the share of DREAMers with a high school degree that choose to attend college in order to obtain permanent residence ($\gamma_2^L = \gamma_2^N = \gamma$).

The top panel in the table (scenarios 1–3) presents the results corresponding exactly to the long-run effects on GDP according to the set of equations in Section 7.3. It is helpful to begin by considering scenario 1, where college enrollment is unaffected by the DREAM Act ($\gamma = 0$). In this case, we find that GDP will increase by 0.05%, which amounts to an overall increase of \$9 billion per year. To provide a more intuitive measure of the size of the effects, it helps to consider that 1.65 million individuals benefit from legalization in our calculations and, out of those, 0.99 million are working in the long-run counterfactual. On the basis of the latter figure, the average long-run increase in GDP per legalized worker results in \$9,104. Because there are no participation effects in this scenario, the short-run change in GDP coincides exactly with the short-run value.

Let us now take into account the increased incentives to attend college (scenario 2). Specifically, we assume $\gamma = 1/2$, that is, we assume that one in two high school graduates with

Scenarios	(1)	(2)	(3)	(4)	(5)
	ΔGDP	ΔGDP	Legalized—All	Legalized—Workers	ΔGDP
	Pct. Change	\$ billions	Millions	Millions	\$ per worker
Legalization					
(1) $\gamma = 0$	0.05	9.0	1.65	0.99	9,104
(2) γ= 0.50	0.08	15.2	1.65	0.99	15,371
(3) γ= 1	0.11	21.3	1.65	0.99	21,519
Removal	-0.42	-81.5	2.93	1.77	-46,061

 Table 7
 Long-run Effects of the DREAM Act on GDP

Notes: Scenarios 1–3 report the long-run gains in GDP associated with passing the DREAM Act when a fraction γ of DREAMers with a high school degree chooses to enroll in college to obtain an associate degree (education level 3). In all scenarios, the new graduates are assumed to work only if they were working in the baseline scenario. GDP amounts in columns 2 and 5 are in 2017 prices. Columns 3 and 4 report the number of individuals that obtain legalization according to our simulation (in millions), with the latter restricting to legalized individuals that are working in the long-run DREAM Act scenario. Column 5 is computed by dividing column 2 by column 4. The last row (Removal scenario) reports the change in GDP associated with removing all DREAMers (undocumented individuals that arrived as children).

conditional status choose to obtain an associate's degree (or 2 years of schooling toward a bachelor's degree). In the short run, there will be two opposing effects on GDP. On the one hand, there is a productivity boost associated with obtaining conditional legal status, as was the case with DACA. However, this positive effect is practically neutralized by a sizable *negative* participation effect driven by the high school graduate DREAMers that leave the workforce to enroll in college. As a result, GDP is practically unaffected in the short run.²⁶ However, over time a sizable positive effect on income would emerge. The long-run effect on GDP reflects a sizable *positive participation effect*: the individuals that left for college (and were initially employed) return to the labor market with enhanced productivity. This leads to a 0.08% increase in GDP (or \$15.2 billion per year), which amounts to \$15,371 per employed legalized individual. Last, scenario 3 considers a more extreme participation effect, where all DREAMers with a high school degree choose to obtain some college education (y = 1). In this case, GDP would increase by 0.11%.

In sum, our analysis implies that passing the DREAM Act will increase the economic contribution of DREAMers that obtain legal status. We estimate that GDP will increase *annually* by an average of 9 to 21 thousand dollars for each worker obtaining legal status. This amount would add to the economic contribution of DREAMers prior to legalization, which can be quantified by comparing GDP in the baseline scenario (prior to legalization) to the level of GDP in a counterfactual where DREAMers are removed from the economy. As reported in the bottom row of Table 7, removal of DREAMers from the workforce would entail a 0.42% reduction in GDP, amounting to \$46,061 per worker. As a result, passing the DREAM Act would increase the *overall* annual contribution of DREAMers to GDP to be around \$60,000 per worker.

7.4.2 Effects of the DREAM Act on wages

Our estimates for the long-run effects on wages are collected in Table 8. Columns 1–3 refer to wage effects pertaining to individuals that did not experience a change in status, that is, documented individuals (who stayed documented) and undocumented individuals that did not benefit from legalization.²⁷ Column 1 reports the wage effects for this population by education–age groups. As expected, the wages of individuals with education level 3 (some college) fall, whereas there is an increase in the wages of individuals in all other education–age groups. However, it is important to note that the magnitudes of the wage effects are very small. Column 3 reports the percent average wages by education level (using the weights reported in column 2). Workers with some college would see their wages fall by 0.22 percent on average, and workers with

²⁶ Even though not reported in Table 7, we can also calculate the short-run effects of legalization on GDP. These effects can be negative for high values of *y*, reflecting the reduction in the workforce when DREAMers with a high-school degree choose to quit their jobs in order to enroll in college. However, this finding could easily be overturned if individuals can simultaneously work and attend college. In fact, this seems to be the case for a large share of immigrant students attending community colleges (Hsin and Ortega 2017).

²⁷ Recall that because of the assumption of perfect substitution in production among workers with the same education and age, the percent change in the wages of groups that did not change documentation status but share the same skills will be identical. Our setup also assumes that natives and legal immigrants with the same age and education are perfect substitutes. To the extent that undocumented workers are closer substitutes for legal immigrant workers than for natives, our predictions will tend to underestimate the wage effects for documented, immigrant workers. However, the small size of the Dreamer population as a fraction of the labor force, and the substantial differences in the education and age distributions between legal and undocumented immigrants imply that the degree of underestimation is probably negligible. Undocumented immigrants are younger and less educated than immigrants with legal status.

					0		
Edu	Age	(1)	(2)	(3)	(4)	(5)	(6)
		Doc	Doc	Doc	Legalized	Legalized	Legalized
		%∆ wage	labor shares	%∆ wage	%∆ wage	labor shares	%∆ wage
HSD	1	0.03	0.24	0.03		0	•
HSD	2	0.03	0.17			0	
HSD	3	0.03	0.20			0	
HSD	4	0.03	0.23			0	
HSD	5	0.03	0.16		•	0	
HSG	1	0.37	0.19	0.16	22.20	0.55	52.37
HSG	2	0.21	0.18		85.09	0.35	
HSG	3	0.1	0.21		115.36	0.09	
HSG	4	0.08	0.26			0	
HSG	5	0.08	0.16			0	
SoCo	1	-0.42	0.22	-0.22	3.56	0.61	15.33
SoCo	2	-0.34	0.22		31.21	0.32	
SoCo	3	-0.15	0.21		45.33	0.07	
SoCo	4	-0.08	0.22			0	
SoCo	5	-0.07	0.14			0	
CoGrad	1	0.03	0.10	0.03	0.03	0.42	0.67
CoGrad	2	0.03	0.25		0.03	0.49	
CoGrad	3	0.02	0.25		7.23	0.09	
CoGrad	4	0.03	0.23			0	
CoGrad	5	0.03	0.17		•	0	

 Table 8
 Wage Effects of DREAM Act. Percent Changes Relative to Baseline

Notes: The table reports long-run percent changes in wages in the DREAM Act scenario. Eligible individuals are required to have a high school diploma in the baseline data. Columns 1–3 refer to documented individuals (Doc), either foreign-born or US-born. Columns 4–6 refer to individuals that obtained legal status through the DREAM Act. Columns 1 and 4 report wage growth (in percent) by education–age. Columns 2 and 5 report the labor shares in the baseline among documented workers (column 2) and among DREAMers (column 5). Columns 3 and 6 report age-weighted average wages by education. A "." denotes a missing value due to the fact that there are no individuals in that education–age-documentation status category. The simulation assumes that a fraction γ = 0.50 of high school graduate DREAMers will obtain some college education (education level 3) to obtain legal permanent residence. Because documented and undocumented workers with the same education and age are perfect substitutes in production, they experience identical education–age-specific wage growth rates. Thus, column 2 can also be applied to undocumented workers that did not obtain legal status.

a high school degree would experience a 0.16 percent increase. At the same time, the wages of individuals at the top and bottom of the education distribution would remain practically unaffected.

Next, we turn to the individuals who obtain legal status (columns 4-6).²⁸ Naturally, the wages of these workers will experience much larger changes. However, we find a great deal of heterogeneity in the size of the wage effects. On the basis of the results in column 6, college graduates that obtain legal status will experience a meager 0.67% average increase in their wages. The reason for this small increase can be traced back to the calibration for the

²⁸ Under our assumptions, only DREAMers with a high school degree are eligible under the DREAM Act.

documented-undocumented productivity gap, which was basically non-existing. In contrast, individuals with some college education that obtain legal status will see their wages increase by an average of 15.33% thanks to the elimination of a substantial undocumented productivity penalty. Yet, our estimates suggest that the largest average wage increase would correspond to high school graduate DREAMers obtaining legalization, with a 52% increase. The reason is that the average individual in this group benefits both from the increase in productivity associated with legal status and from rewards to the increased educational attainment.

Lastly, it is important to keep in mind that, on the basis of our analysis, the wage increases experienced by individuals that obtain legal status largely reflect increases in productivity arising from enhanced education as well as improved access to jobs where workers can make a better use of their skills. To the extent that the wage increase is productivity driven, rather than due to increased bargaining power on the part of the worker, employers' labor costs need not be affected.

8 Conclusions

This paper has developed a simple general equilibrium model that can be used to quantify the economic gains from legalizing undocumented workers that arrived in the United States as children. Our model extends the framework proposed by Edwards and Ortega (2017) by considering a variety of participation and education effects. We use the model to simulate the effects of temporary legalization as implemented through the DACA program, as well as the effects of offering a track to permanent residence through the 2017 Senate version of the DREAM Act.

At some level, both modes of legalization share the feature that they are likely to increase the productivity of workers who obtain legal status because of the improved labor market opportunities. However, there are important differences between the two modes of legalization, stemming from participation effects of different sign and magnitude. DACA entails a positive participation effect, driven by the many undocumented college students that dropped out in order to take advantage of the improved labor market opportunities. While this effect increases the short-run effect of DACA on GDP, it may entail a cost in the long run given that it is unlikely that these individuals return to college in the future.

In comparison, the DREAM Act entails a *negative* participation effect in the short run because it is likely to induce some undocumented high school graduates that were initially employed to quit their jobs and enroll in college in order to obtain permanent residence.²⁹ In contrast, the long-run effect on GDP can be rather large when the new college graduates return to the workforce. We estimate that the long-run increase in GDP will range between 0.05% and 0.11%. We have also analyzed the wage effects of legalization under DACA and the DREAM Act. Because DREAMers are only a small fraction of the population, legalization has very small effects on the wages of native workers. In contrast, the wages of most individuals gaining legal status will increase substantially, with the largest increases being experienced by DREAMers that increase their educational attainment in order to qualify for legalization.

²⁹ Under some parameter values, this effect is large enough that it may overshadow the productivity gains associated with legal status.

Our analysis has not explicitly considered tax implications such as the role of payroll taxes. At some level, it is possible that payroll tax considerations might compel employers to shift some demand toward workers who originally were documented. This provides an alternative explanation, in addition to the basic complementarity story, for why the wages of documented workers go up when DREAMers become authorized. However, it is also worth noting that many undocumented workers have been hired using someone else's social security number (David Brown et al. 2013). In those cases, employers are already paying payroll taxes and legalization will not affect the employer cost of labor.

Providing legal status to DREAMers could entail a fiscal cost because of increased access to public services (Cascio and Lewis 2019). According to the Congressional Budget Office (memo *S. 1615, Dream Act of 2017*), passing the DREAM Act will increase government spending by \$2.68 billion per year. Our analysis predicts that passing the DREAM Act would boost GDP by about \$15.2 billion, which will increase tax revenue. Again based on United States Congressional Budget Office (CBO) forecasts, each dollar increase in GDP translates into 18 cents of federal tax revenue. Thus, our estimated GDP gains translate into an increase in federal tax revenue of \$2.73 billion per year. Hence, passing the DREAM Act is likely to be essentially budget-neutral.

We close by noting that the GDP effects of the DREAM Act could be substantially larger than the estimates presented here. The reason is that we have limited our analysis to DREAMers that have completed high school. However, one would expect that passing the DREAM Act is likely to encourage many DREAMers that had not completed high school to go back to school in order to become eligible for legalization. Our framework could be extended in order to incorporate this additional educational response to the eligibility requirements of the DREAM Act.

Declarations

Ethics approval and consent to participate

No human subjects are involved in the research. The data are anonymous and the research design was approved by the custodian of the data, the Center for Migration Studies (New York).

Consent for publication

Not applicable.

Availability of data and material

The data that support the findings of this study are available from the Center for Migration Studies (New York) but restrictions apply to the availability of these data, which were used under license for the current study, and so are not publicly available. Data are however available from the authors upon reasonable request and with permission of the Center for Migration Studies (New York). The data used is a version of the ACS that includes an imputation for undocumented status. Except for that variable, all other variables can be obtained from the public version of the dataset on IPUMS.

Competing interests

The authors declare that they have no competing interests.

Funding

The research was funded by the MacArthur Foundation (grant 15-1643). The funding body did not influence the design, analysis, or interpretation of the findings.

Authors' contributions

The three authors are jointly responsible for the research analysis. RE was the leading author in acquiring and processing the data. FO was responsible for the coding of the equations using in the simulation. The three

authors were jointly responsible for the interpretation of the findings. AH and FO are responsible for drafting of the text. All authors have approved the submitted version and agree both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.

Acknowledgments

We thank Catalina Amuedo-Dorantes, Maryann Bertrand, Julie Berry Cullen, Lorenzo Caliendo, Julen Esteban-Pretel, Chinhui Juhn, Fernando Lozano, Joel Machado, Joan Monras, Nuria Rodriguez-Planas, Kevin Shih, Thom Thurston, Marcos Vera-Hernandez, Bob Warren, and all seminar participants at San Diego State University, the Philadelphia Fed, Pomona College, CReAM-UCL, and participants at Columbia University's workshop on *Shaping the Future of DACA* for helpful comments.

References

- Abrego, Leisy J (2011): Legal Consciousness of Undocumented Latinos: Fear and Stigma as Barriers to Claims-Making for First-and 1.5-Generation Immigrants. Law & Society Review 45(2), 337-370.
- Amuedo-Dorantes, Catalina; Francisca Antman (2017): Schooling and Labor Market Effects of Temporary Authorization: Evidence from DACA. Journal of Population Economics 30(1), 339-373.
- Baker, Bryan C.; Nancy Rytina (2012): Estimates of the Unauthorized Immigrant Population Residing in the United States: January 2012. Washington, DC: Office of Immigration Statistics, Policy Directorate, US Department of Homeland Security. http://www.dhs.gov/sites/default/les/publications/ois_ill_ pe_2012_2.pdf, 2013.
- Borjas, George J (2016): The Labor Supply of Undocumented Immigrants. NBER Working Papers 22102, National Bureau of Economic Research, Inc.
- Borjas, George (2003): The Labor Demand Curve is Downward Sloping: Reexamining the Impact of Immigration on the Labor Market. Quarterly Journal of Economics 118(4), 1335-1375.
- David Brown, J.; Julie L. Hotchkiss; Myriam Quispe-Agnoli (2013): Does Employing Undocumented Workers Give Firms A Competitive Advantage? Journal of Regional Science 53(1), 158-170.
- Card, David; Thomas Lemieux (2001): Can Falling Supply Explain the Rising Return to College for Younger Men? A Cohort-Based Analysis. The Quarterly Journal of Economics 116(2), 705-746.
- Casarico, Alessandra; Giovanni Facchini; Tommaso Frattini (2018). What Drives the Legalization of Immigrants? Evidence from IRCA. Regional Science and Urban Economics 70(C), 258-273.
- Cascio, Elizabeth U.; Ethan G. Lewis (2019): Distributing the Green (Cards): Permanent Residency and Personal Income Taxes after the Immigration Reform and Control act of 1986. Journal of Public Economics 172, 135-150.
- Center for Migration Studies. (2014): Estimates of the Unauthorized Population. Dataset based on the Augmented American Community Survey Datafiles Hosted by IPUMS (Integrated Public Use Microdata Series), 2010 to 2014.
- Chassambouli, Andri; Giovanni Peri (2015): The Labor Market Effects of Reducing the Number of Illegal Immigrants. Review of Economic Dynamics 18(4), 792-821.
- Cho, Esther Yoona (2017): Revisiting Ethnic Niches: A Comparative Analysis of the Labor Market Experiences of Asian and Latino Undocumented Young Adults. RSF.
- Devillanova, Carlo; Francesco Fasani; Tommaso Frattini (2018): Employment of Undocumented Immigrants and the Prospect of Legal Status: Evidence from an Amnesty Program. ILR Review 71(4), 853-881.
- Edwards, Ryan; Francesc Ortega (2017): The Economic Contribution of Unauthorized Workers: An Industry Analysis. Regional Science and Urban Economics 67, 119-134.
- Fasani, Francesco (2018): Immigrant Crime and Legal Status: Evidence from Repeated Amnesty Programs. Journal of Economic Geography 18(4), 887-914.
- Gleeson, Shannon; Roberto G. Gonzales (2012): When do Papers Matter? An Institutional Analysis of Undocumented Life in the United States. International Migration 50(4), 1-19.
- Goldin, Claudia; Lawrence F. Katz (2008): Transitions: Career and Family Life Cycles of the Educational Elite. American Economic Review 98(2), 363-369.
- Gonzales, Roberto G. (2011): Learning to be Illegal: Undocumented Youth and Shifting Legal Contexts in the Transition to Adulthood. American Sociological Review 76(4), 602-619.
- Hainmueller, Jens; Duncan Lawrence; Linna Martén; Bernard Black; Lucila Figueroa; Michael Hotard; Tomás R Jiménez; Fernando Mendoza; Maria I Rodriguez; Jonas J Swartz, et al. (2017): Protecting Unauthorized Immigrant Mothers Improves their Children's Mental Health. Science 357(6355), 1041-1044.

- Hall, Matthew; Emily Greenman; George Farkas (2010): Legal Status and Wage Disparities for Mexican Immigrants. Social Forces 89(2), 491-513.
- Hall, Matthew; Emily Greenman (2015): The Occupational Cost of Being Illegal in the United States: Legal Status, Job Hazards, and Compensating Differentials. International Migration Review 49(2), 406-442.
- Hirsch, Boris; Elke J. Jahn (2015): Is There Monopsonistic Discrimination against Immigrants? ILR Review 68(3), 501-528.
- Hotchkiss, Julie L.; Myriam Quispe-Agnoli (2009): Employer Monopsony Power in the Labor Market for Undocumented Workers. FRB Atlanta Working Paper 2009-14, Federal Reserve Bank of Atlanta.
- Hsin, Amy; Francesc Ortega (2017): The Effects of Deferred Action for Childhood Arrivals on the Educational Outcomes of Undocumented Students: Evidence from A Large Public University. IZA Discussion Papers 11078, Institute for the Study of Labor (IZA).
- Kossoudji, Sherrie A.; Deborah A. Cobb-Clark (2002): Coming out of the Shadows: Learning about Legal Status and Wages from the Legalized Population. Journal of Labor Economics 20(3), 598-628.
- Kuka, Elira; Na'ama Shenhav; Kevin Shih (2018): Do Human Capital Decisions Respond to the Returns to Education? Evidence from DACA. NBER Working Papers 24315, National Bureau of Economic Research, Inc.
- Lofstrom, Magnus; Laura Hill; Joseph Hayes (2013): Wage and Mobility Effects of Legalization: Evidence From the New Immigrant Survey. Journal of Regional Science 53(1), 171-197.
- Lozano, Fernando A.; Todd A. Sorensen (2011): The Labor Market Value to Legal Status. IZA Discussion Papers 5492, Institute for the Study of Labor (IZA).
- Lynch, Robert; Patric Oakford (2013): The Economic Effects of Granting Legal Status and Citizenship to Undocumented Immigrants. Center for American Progress, March 20 2013.
- Machado, Joel (2017): Dealing with Undocumented Immigrants: The Welfare Effects of Amnesties and Deportations. Journal of Demographic Economics 83(4), 445-492.
- Manacorda, Marco; Alan Manning; Jonathan Wadsworth (2012): The Impact of Immigration on the Structure of Wages: Theory and Evidence from Britain. Journal of the European Economic Association 10(1), 120-151.
- Monras, Joan; Javier Vázquez-Grenno; Ferran Elias Moreno (2017): Understanding the Effects of Legalizing Undocumented Immigrants. IZA Discussion Papers 10687, Institute for the Study of Labor (IZA).
- Orrenius, Pia M.; Madeline Zavodny (2015): The Impact of e-Verify Mandates on Labor Market Outcomes. Southern Economic Journal 81(4), 947-959.
- **Orrenius, Pia M.; Madeline Zavodny** (2016): Irregular Immigration in the European Union. Working Papers 1603, Federal Reserve Bank of Dallas.
- Ortega, Francesc; Amy Hsin (2018): Occupational Barriers and the Labor Market Penalty from Lack of Legal Status. IZA Discussion Papers 11680, Institute for the Study of Labor (IZA).
- Ottaviano, Gianmarco I. P.; Giovanni Peri (2012): Rethinking the Effect of Immigration on Wages. Journal of the European Economic Association 10(1), 152-197.
- Passel, Jeffrey S.; Rebecca L. Clark (1998): Immigrants in New York: Their Legal Status, Incomes and Taxes. Washington, DC: Urban Institute.
- Passel, Jeffrey S.; D'Vera Cohn (2015): Share of Unauthorized Immigrant Workers in Production, Construction Jobs Falls Since 2007: In States, Hospitality, Manufacturing and Construction are Top Industries. Washington, D.C.: Pew Research Center.
- Pastor, Manuel; Justin Scoggins (2016): Estimating the Eligible-to-Naturalize Population. Center for the Study of Immigrant Integration.
- Pope, Nolan G. (2016): The Effects of DACAmentation: The Impact of Deferred Action for Childhood Arrivals on Unauthorized Immigrants. Journal of Public Economics 143(C), 98-114.
- Van Hook, Jennifer; James D. Bachmeier; Donna L. Coffman; Ofer Harel (2015): Can we Spin Straw into Gold? An Evaluation of Immigrant Legal Status Imputation Approaches. Demography 52(1), 329-354.
- Vogel, Dita; Vesela Kovacheva; Hannah Prescott (2011): The Size of the Irregular Migrant Population in the European Union Counting the Uncountable? International Migration 49, 78-96.
- Warren, Robert; John Robert Warren (2013): Unauthorized Immigration to the United States: Annual Estimates and Components of Change, by State, 1990 to 2010. International Migration Review 47(2), 296-329.
- Warren, Robert (2014); Democratizing Data about Unauthorized Residents in the United States: Estimates and Public-Use Data, 2010 to 2013. Journal on Migration and Human Security 2(4), 305-328.